Author's response to reviews

Title: Rapid detection of dermatophytes and Candida albicans in onychomycosis specimens by an oligonucleotide array

Authors:

Huan Wen Han (p8896116@mail.ncku.edu.tw)
Mark Ming-Long Hsu (a441224@mail.ncku.edu.tw)
Jong Soo Choi (jschoi@med.yu.ac.kr)
Chao-Kai Hsu (kylehsu@mail.ncku.edu.tw)
Hsin Yi Hsieh (toheart0824@yahoo.com.tw)
Hsin Chieh Li (b8803126@stmail.cgu.edu.tw)
Hsien Chang Chang (hcchang@mail.ncku.edu.tw)
Tsung Chain Chang (tsungcha@mail.ncku.edu.tw)

Version: 3  
Date: 16 October 2014

Author's response to reviews: see over
Dear Dr. Harris:

Thank you for processing our manuscript entitled “Rapid detection of dermatophytes and Candida albicans in onychomycosis specimens by an oligonucleotide array” (MS: 1307216199106946). We appreciate the valuable comments from the three reviewers. Their comments/suggestions help to improve the manuscript quality. The manuscript has been modified according to the comments of these reviewers. We have made an item by item response to their comments. We also ensure that our revised manuscript conforms to the BMC journal style. Thank you again for effort to process this revised article.

Best regards,

Tsung Chain Chang, PhD
Professor, Department of Medical Laboratory Science and Biotechnology
College of Medicine
National Cheng Kung University
1 University Road
701 Tainan, Taiwan
Tel: 886-6-2353535 ext 5790
Fax: 886-6-2363956
e-mail: tsungcha@mail.ncku.edu.tw